Author’s response to reviews

Title: New variant identified in major susceptibility locus to Tuberculosis on chromosomal region 8q12-q13 in Moroccan population: a case control study

Authors:
mounia qrafli (mounia.qrafli@gmail.com)
Imane Asekkaj (imane.asekkaj@gmail.com)
Jamal eddine Bourkadi (jebourkadi@yahoo.com)
Rajae elaouad (elaouadrajae@gmail.com)
khalid SADKI (khalidsadki1@gmail.com)

Version: 1 Date: 26 Jul 2017

Author’s response to reviews:

* There are a number of language errors throughout the manuscript and as a result we need to request that you copyedit your manuscript before we can proceed.

-------

As requested, the present manuscript was copyedited by Americain Journal Experts